## Changes in marital sorting: Theory and evidence from the US

Pierre-André Chiappori\* Monica Costa Dias<sup>†</sup> Costas Meghir<sup>‡</sup>

Hanzhe Zhang<sup>§</sup>

November 26,  $2024^{\P}$ 

### Abstract

Positive assortative matching refers to the tendency of individuals with similar characteristics to form partnerships. Measuring the extent to which assortative matching differs between two economies is challenging when the marginal distributions of the characteristic along which sorting takes place (e.g., education) change for either or both sexes. We show how the use of different measures can generate different conclusions. We provide axiomatic characterization for measures such as the odds ratio, normalized trace, and likelihood ratio, and provide a structural economic interpretation of the odds ratio. We then use our approach to consider how marital sorting by education changed between the 1950s and the 1970s cohort, for which both educational attainment and returns in the labor market changed substantially.

Keywords: sorting, homogamy, educational attainment, marriage markets **JEL**: C78, D1, J1

<sup>\*</sup>Columbia University, pc2167@columbia.edu

<sup>&</sup>lt;sup>†</sup>University of Bristol and Institute for Fiscal Studies, monica.costa-dias@bristol.ac.uk

<sup>&</sup>lt;sup>‡</sup>Yale University, NBER, IFS, IZA, CEPR, IFAU, c.meghir@yale.edu

<sup>&</sup>lt;sup>§</sup>Michigan State University, hanzhe@msu.edu

<sup>&</sup>lt;sup>¶</sup>This paper subsumes "Changes in Assortative Matching: Theory and Evidence for the US" (Chiappori, Costa Dias, and Meghir, 2020) and "Axiomatic Measures of Assortative Matching" (Zhang, 2024). We are grateful to the editor James Heckman for his detailed comments and guidance, as well as to four anonymous referees. We thank Dan Anderberg, Hector Chade, Willy Chen, Juanna Joensen, Maciej Kotowski, Jingfeng Lu, Magne Mogstad, Antonio Nicolò, Bernard Salanié, Edward Schlee, Mark Whitemeyer, Basit Zafar, Mu Zhang and Jin Zhou. Costa Dias gratefully acknowledges financial support from the Economic and Social Research Council through the ESRC Centre for the Microeconomic Analysis of Public Policy at IFS (reference ES/M010147/1) and ESRC grants ES/K00624X/1 and ES/N015304/1. Meghir gratefully acknowledges financial support from the Cowles Foundation and the Institution for Social and Policy Studies at Yale. Zhang gratefully acknowledges financial support from National Science Foundation and Michigan State University Delia Koo Endowment Fund. All errors are our own.

## 1 Introduction

The study of sorting in the marriage market has recently attracted renewed attention. The degree of homogamy in marriage—defined as people's tendency to "marry their own"—has important consequences for family inequalities and intergenerational transmission of human capital. Therefore, it is surprising that various studies on this topic have not reached a consensus on the evolution of homogamy in recent years, and even on how best to measure homogamy (e.g., Fernández and Rogerson, 2001; Mare and Schwartz, 2005; Greenwood et al., 2014; Siow, 2015; Chiappori, Salanié, and Weiss, 2017; Eika, Mogstad, and Zafar, 2019; Ciscato and Weber, 2020; Gihleb and Lang, 2020; Hou et al., 2022).

The goal of this paper is to understand why different approaches to the same problem, using similar or even the same data, can reach opposite conclusions and, more generally, to clarify the theoretical issues underlying the choice of a particular measure of assortativeness. Our analysis considers two populations, men and women, sorting in marriage according to a characteristic, say education. Whenever the marginal distributions change—e.g., women's average education increases—matching patterns will change. The main problem faced by any measure of assortativeness is to disentangle the mechanical effects of such variations in the marginal distributions from deeper changes in the matching structure itself—for instance, originating from changes in the gain generated by assortativeness along that characteristic. The latter represents what one would call "changes in the assortativeness." Existing studies propose various measures—indices and rankings—to capture the assortativeness and the changes therein. They measure assortativeness in different ways and may therefore generate divergent conclusions.

Rather than starting with specific measures and justifying them with selected desirable properties they satisfy, we take an *axiomatic approach*: We start with a set of properties that the measures should satisfy and establish how appropriate the measures are for capturing changes in assortative matching through the lenses of these properties. We consider three sets of basic axioms: (i) axioms of *invariance* to specify when two matching patterns are equally assortative, (ii) axioms of *monotonicity* to specify when one matching pattern is more assortative than another, and (iii) axioms of *homogamy* to specify matching patterns that are most assortative.<sup>1</sup> Namely, invariance axioms are *scale invariance* (invariance to scaling the market), *side invariance* (invariance to swapping sides), and *type invariance* (invariance to swapping types). The two notions of monotonicity we consider are (i) *marginal monotonicity*, ranking two sets of matches of the same marginal distributions, and (ii) *diagonal* and *off-diagonal monotonicity* (assortativeness increases when more like types match and decreases when more unlike types match). Finally, we introduce two definitions of the most assortative matching: *full homogamy* (every pair in the market is homogamous) and *maximum homogamy* (the maximum number of homogamous).

While some commonly used measures, such as the likelihood ratio, fail to satisfy one or more of the basic axioms, the *odds ratio* (the ratio of college and noncollege individuals' odds to match with the same type and with a different type), the *normalized trace* (the proportion of like types), and the *minimum distance* (the combination of sorting patterns under random matching and maximum positive assortative matching that best fits the empirical observation) satisfy all basic axioms. Since these are genuinely different measures rather than monotonic transformations of each other, they may point in opposite directions when used to investigate changes in assortativeness between two matching patterns. Their differences are empirically relevant. We show estimates of changing marital sorting patterns in the US where one finds instances of these measures reaching opposite conclusions. To distinguish some of these measures, we additionally define a set of *characterization axioms*.

In two-type markets, the complete ranking induced by the odds ratio is the unique one that satisfies the basic axioms plus *marginal independence* (Edwards, 1963) (Theorem 1). Under marginal independence, only the odds of marrying different types of spouses matter for

<sup>&</sup>lt;sup>1</sup>In this paper, we use axioms and properties interchangeably.

the measure of assortativeness, not the marginal distributions of the types. All indices that are monotonic transformations of the odds ratio (e.g., Yule's Q, Yule's Y, and log odds ratio) provide the same assortativeness ordering. We also provide a structural economic interpretation of the odds ratio: It estimates the average gain per couple from assortative matching compared to nonassortative matching within the Choo and Siow (2006) transferable-utilities matching framework.

We also define *decomposability* axioms: The assortativeness measure for any matching in a market is a weighted average or weighted sum of the measures for the matching patterns in the submarkets decomposed from the market. The *normalized trace*—the proportion of pairs of like types—is the unique index, up to positive affine transformation, that satisfies the basic axioms plus *population decomposability*, i.e., the weight to average is population size (Theorem 2). For comparison, we also provide an axiomatic characterization of the *aggregate likelihood ratio* (used, for example, in Eika, Mogstad, and Zafar, 2019)—the excess likelihood of like types pairing up relative to the hypothetical likelihood under random matching. It is the unique measure, up to positive multiplication, that satisfies selected basic axioms plus *random decomposability*, i.e., the weight to sum is the hypothetical proportion of pairs of like types under random matching (Theorem 3).

For markets with more than two types, the indices such as normalized trace and likelihood ratio can be extended to provide measures of assortativeness.<sup>2</sup> In practice, there are 14 detailed education categories in the US Census, so *robustness to categorization* is an additional desired property: Comparisons of assortativeness should be robust to how education is categorized into two or several groups. We provide an impossibility result: No total preorder satisfies marginal monotonicity (the weaker monotonicity axiom) and robustness to categorization (Theorem 4). As a result, we must resort to partial rankings to evaluate assortativeness of multi-type matching patterns.

The continued attention in the literature to how assortativeness on education has evolved

 $<sup>^{2}</sup>$ They can also be modified to compare markets with singles and one-sided (e.g., homosexual) matches (Zhang, 2024).

in the recent past has been fueled by contradictory findings. We revisit this issue using our axiomatic approach, studying changes in assortativeness on education in the US, across cohorts born in the 1950s and 1970s. We contribute to that discussion by arguing that ambiguity can sometimes be resolved by selecting indices that have desirable properties, and by studying local (submarket) changes in assortativeness rather than global ones, which are hard to interpret. The indices that satisfy our basic axioms consistently show that assortativeness by education has increased at the top of the education distribution, but has decreased at the very top, between college graduates and post graduates. We also find that the type-specific and aggregate likelihood ratios—which fail to satisfy a number of the basic axioms of invariance, monotonicity, and homogamy—frequently contradict the findings of other indices and of each other.

The rest of the paper is organized as follows. Section 2 sets up the model and discusses special matching patterns and commonly used measures. Section 3 lays out the basic axioms. Section 4 discusses the axiomatic characterization results for the odds ratio, normalized trace, and aggregate likelihood ratio. Section 5 discusses the structural interpretation of the odds ratio. Section 6 discusses the theoretical results with more than two types. Section 7 provides empirical results on marital sorting in the US, and Section 8 concludes.

## 2 Model, matchings, and measures

### 2.1 Model and objective

Each man and woman possesses a trait (e.g., college education or any other observable psychological, biological, or socioeconomic trait). We start with the setting where a trait is one of two types, e.g.,  $\theta_1$  and  $\theta_2$ . For expositional ease and alignment with our primary empirical application, we refer to the two traits as college-educated and noncollege-educated.

Consider the matching between men and women described by matrix M = (a, b, c, d):

$$m \setminus w \qquad \theta_1 \text{ (college)} \quad \theta_2 \text{ (noncollege)}$$
$$M = \begin{array}{cc} \theta_1 \text{ (college)} & a & b \\ \theta_2 \text{ (noncollege)} & c & d \end{array}$$

A cell describes the mass of pairs between a specific combination of types of men and women. To recover the marginal distribution, i.e., mass of (matched) individuals of a specific gender and type, we sum a column or a row. For example, there is mass a + b of college men. The marginal distribution for men is then described by (a + b, c + d) and that for women is (a + c, b + d). Assume a full support of types on both sides of the market: (a + b)(a + c)(b + d)(c + d) > 0. Let  $|M| \equiv a + b + c + d$  denote its population size.

We define *positive matching matrices* M = (a, b, c, d), i.e., matrices such that each of the entries a, b, c, and d is strictly positive.<sup>3</sup> We denote them by M > 0, where bold **0** denotes the zero matrix, the 2 × 2 zero matrix in this context.

**Objective.** Our aim is to define a set of axioms that a measure of assortative matching should reasonably satisfy and to use these to distinguish among different measures, which often lead to contradictory results. A measure may be an indexing, a complete ranking, or a partial ranking of all matching patterns.

Formally, let  $\mathcal{M} = \mathbb{R}^4_+ \setminus \{0\}$  denote the entire collection of possible matching patterns in two-type markets. A measure of assortativeness may be (i) an *index*  $I : \mathcal{M} \to \mathbb{R}_+$ ; (ii) a *total preorder* (i.e., a complete ranking) on  $\mathcal{M}$ , a binary relation that satisfies totality  $(M \succeq M')$  or  $M' \succeq M$ ) and transitivity  $(M \succeq M')$  and  $M' \succeq M'')$  imply  $M \succeq M'')$ ; or (iii) a *preorder* (i.e., a partial or complete ranking) on  $\mathcal{M}$ , a binary relation that satisfies reflexivity  $(M \succeq M)$ and transitivity.<sup>4</sup> Any index is associated with a total preorder, so studying the ordinal properties of an index is akin to studying the properties of the total preorder it induces.

<sup>&</sup>lt;sup>3</sup>Note that positive matching matrices are distinct from positive definite matrices.

<sup>&</sup>lt;sup>4</sup>We consider preorders rather than orders (which satisfy the additional condition of antisymmetry:  $M \succeq M'$  and  $M' \succeq M$  imply M = M') because there may be equally assortative matching patterns.

### 2.2 Special matching patterns

Fully and maximally assortative matching. When a person is matched with a partner of the same type with probability one, i.e., b = c = 0, we call such a matching *fully positive-assortative*. However, fully positive-assortative matching is feasible only when men and women have the same distribution of education. More generally, we call a matching *maximally positive-assortative* if there is a maximum mass of pairs of like types, holding marginal distributions fixed: bc = 0 (that is, b = 0 or c = 0 or both). Say, all individuals on the short side of the educated category (men in our case) marry an educated partner (i.e., b = 0), but not every educated individual on the long side (women) does so (i.e., c > 0); intuitively, the only reason that we observe "mixed" couples is the lack of educated men. Every market has a unique maximally positive-assortative matching.

In the two-by-two case, we can analogously define *fully negative-assortative* matching as one in which all individuals are matched with a partner of a different type with probability one, i.e., a = d = 0; and *maximally negative-assortative* (or equivalently, *minimally positiveassortative*) matching as one in which there is a maximal mass of pairs of unlike types (or equivalently, in two-by-two markets, minimal mass of pairs of like types), i.e., ad = 0.

Note that fully (positive or negative) assortative matching is maximally assortative, but maximally assortative matching is not necessarily fully assortative. Most—not all—of the preorders in use have maximally positive-assortative matching matrices as maximum elements of  $\mathcal{M}$ .

Positive and negative assortative matching. One way of defining positive and negative assortative matching is to compare to the random matching benchmark. We say that matching M is a *positive assortative matching (PAM)* if the mass of couples with equal education (the "diagonal" of matrix M) is strictly larger than what would obtain under random matching. Under random matching, the mass of couples in which both spouses are college-educated is  $|M| \cdot [(a+b)(a+c)/|M|^2]$ . Then we have PAM if and only if

$$a(a+b+c+d) > (a+b)(a+c),$$
(1)

or equivalently,

$$ad > bc.$$
 (2)

This inequality also implies that more noncollege individuals marry each other than would be predicted by random matching.<sup>5</sup> In other words, PAM arises when extra forces generate more matches between equally educated people than would happen for random reasons. *Negative assortative matching (NAM)* in two-type markets can be analogously defined: ad < bc.

### 2.3 Existing measures and their differences

### 2.3.1 Existing measures

We now review some of the most commonly used measures in the literature. Many are measuring the extent to which the left-hand side is larger than the right-hand side in equations (1) and (2).

Odds ratios. The odds ratio is probably the most widely used index:<sup>6</sup>

$$I_O(M) = \begin{cases} \frac{ad}{bc} & \text{if } b \neq 0 \text{ and } c \neq 0, \\ +\infty & \text{if } bc = 0. \end{cases}$$

The index ranges from 0 to  $+\infty$ . A few monotonic transformations of the odds ratio are used in the literature. They all yield the same preorder of assortativeness. Log odds ratio is

<sup>&</sup>lt;sup>5</sup>Mathematically,  $d > (d+b)(d+c)/|M| \Leftrightarrow d(a+b+c+d) > (d+b)(d+c) \Leftrightarrow d(a+b) > b(d+c) \Leftrightarrow ad > bc$ . The inequality ad > bc also implies a+d > [(a+b)(a+c) + (d+b)(d+c)]/(a+b+c+d).

<sup>&</sup>lt;sup>6</sup>The odds ratio is popular in the demographic literature, as it can be directly derived from the loglinear approach; see, for instance, Mare (2001), Mare and Schwartz (2005), and Bouchet-Valat (2014). In economics, it was used by Siow (2015) ("local odds ratio"), Chiappori, Salanié, and Weiss (2017), Chiappori et al. (2020), and Ciscato and Weber (2020), among many others.

 $I_o(M) \equiv \ln I_O(M)$ . Yule's Q or coefficient of association (Yule, 1900) is  $I_Q(M) \equiv \frac{I_O(M)-1}{I_O(M)+1}$ . This is +1 when PAM, -1 when NAM, and 0 when random matching (uncorrelated). Yule's Y or coefficient of colligation (Yule, 1912) is  $I_Y(M) \equiv \frac{\sqrt{I_O(M)}-1}{\sqrt{I_O(M)}+1}$ .

Likelihood ratios. *Type-specific likelihood ratio*, used by Eika, Mogstad, and Zafar (2019), for instance, measures marital sorting between men and women of the same type by the ratio of the actual probability of matching relative to what would occur at random:<sup>7</sup>

$$I_{L1}(M) \equiv \frac{\text{observed matches } \theta_1 \theta_1}{\text{random matching baseline}} = \frac{a}{|M|} / \frac{a+b}{|M|} \frac{a+c}{|M|} = \frac{a(a+b+c+d)}{(a+b)(a+c)}$$

The likelihood ratio for population type  $\theta_1$  compares the realized mass of college-college pairs with the benchmark of the hypothetical mass of such pairs under random matching. The likelihood ratio for population type  $\theta_2$  is analogously defined:

$$I_{L2}(M) \equiv \frac{\text{observed matches } \theta_2 \theta_2}{\text{random matching baseline}} = \frac{d}{|M|} \left/ \frac{d+b}{|M|} \frac{d+c}{|M|} = \frac{d(a+b+c+d)}{(d+b)(d+c)}.$$

One issue with the type-dependent likelihood ratio is that it requires the choice of a type as a benchmark; in other words, it fails *type invariance*, which we will define later on. Our empirical results show that the likelihood ratios based on different types yield opposite conclusions on the direction of change in educational homogamy in the US.

The *aggregate likelihood ratio* is a weighted average of the type-specific likelihood ratios, in which the weight on each type-specific likelihood ratio is the expected mass of pairs of

<sup>&</sup>lt;sup>7</sup>Ciscato, Galichon, and Goussé (2020) call this the homogamy rate. It is used to study sorting on career ambition (Almar et al., 2023) and on childhood family income percentile (Binder et al., 2023).

like types under random matching.

$$I_{L}(M) \equiv \frac{(a+b)(a+c)}{(a+b)(a+c) + (d+b)(d+c)} I_{L1}(M) + \frac{(d+b)(d+c)}{(a+b)(a+c) + (d+b)(d+c)} I_{L2}(M)$$
  
=  $\frac{\text{observed matches } \theta_{1}\theta_{1} \text{ and } \theta_{2}\theta_{2}}{\text{random matching baseline}} = \frac{a+d}{|M|} \Big/ \left(\frac{a+b}{|M|}\frac{a+c}{|M|} + \frac{d+b}{|M|}\frac{d+c}{|M|}\right).$ 

Simplified, this is the ratio between the observed mass of couples of all like types and the counterfactual mass of pairs if individuals were to match randomly.

**Normalized trace.** A simple index that turns out to have nice properties is what we call the *normalized trace*.<sup>8</sup> It equals 1 if the matching is maximum PAM, equals 0 if the matching is minimum PAM, and equals the proportion of like types in the market otherwise:

$$I_{tr}(M) = \begin{cases} 1 & \text{if } bc = 0, \\ \frac{a+d}{a+b+c+d} \in (0,1) & \text{if } abcd \neq 0 \\ 0 & \text{if } ad = 0. \end{cases}$$

Minimum distance. In the *minimum distance* approach of Fernández and Rogerson (2001), Abbott et al. (2019), and Wu and Zhang (2021), one constructs the convex combination of two extreme cases—random matching and maximum PAM—that minimizes the distance with the matching under consideration and defines the weight of the maximum PAM component as the index. It coincides with the *perfect-random normalization* of Liu and Lu (2006) and Shen (2020) in two-type markets and is equal to

$$I_{MD}(M) = \frac{ad - bc}{(a + \min\{b, c\})(d + \min\{b, c\})}.$$

<sup>&</sup>lt;sup>8</sup>Using the normalized trace, Cheremukhin, Restrepo-Echavarria, and Tutino (2024) study marital sorting in the US, and Li (2024) and Li and Derdenger (2024) study matching between students and colleges.

**Correlation.** Another natural index is the correlation between wife's and husband's education, each considered as a Bernoulli random variable taking the value  $\theta_1$  with probability  $\frac{a+b}{|M|}$  (resp.,  $\frac{a+c}{|M|}$ ) and  $\theta_2$  with probability  $\frac{c+d}{|M|}$  (resp.,  $\frac{b+d}{|M|}$ ):

$$I_{Corr}\left(M\right) = \frac{ad - bc}{\sqrt{\left(a + b\right)\left(c + d\right)\left(a + c\right)\left(b + d\right)}}.$$

This has been used in various contributions (e.g., Greenwood, Guner, and Knowles, 2003; Greenwood et al., 2014), either explicitly or through a linear regression framework. In our  $2 \times 2$  case, the correlation index also coincides with Spearman's rank correlation, which exploits the natural ranking of education levels (i.e., college > noncollege). Equivalently, one can consider  $\chi^2(M) = [I_{Corr}(M)]^2$ .

### 2.3.2 Measuring changes in sorting with alternative measures: An example

As we show in the empirical section, the various measures can imply opposite results with respect to changes in sorting, an issue that has motivated this paper. We provide a graphical illustration of this difference by comparing the odds ratio and the aggregate likelihood ratio, before we analyze the various measures in relation to a set of axioms.

For the purposes of our example, fix a reference matching, M = (0.2, 0.1, 0.1, 0.6) and restrict attention to matching matrices M' in which |M'|=1 and b = c = (1 - a - d)/2. The red curve in Figure 1 illustrates what we call the *iso-assortative curve* of M— the collection of matrices that are equally assortative as M—under the odds ratio. Similarly, the white curve is the iso-assortative curve of M under the aggregate likelihood ratio.

Thus, the orange (yellow) area is the set of matching matrices where sorting is more positive (less positive) assortative than M according to *both* measures. The purple areas are the set of matrices classified as less assortative than M by the aggregate likelihood ratio, but more assortative according to the odds ratio. Notably, the fully assortative matching matrices—a = 1 and d = 0 or a = 0 and d = 1—and the matching matrices close to them



Figure 1: Iso-assortative curves and conflicting region

are deemed strictly less assortative than M under the aggregate likelihood ratio. Finally, the matrices in the green regions are more assortative than M under the aggregate likelihood ratio, and less so under the odds ratio. These differences reflect the way that each of the measures accounts for changes in the marginal distributions.<sup>9</sup>

## 3 Axioms of invariance, monotonicity, and homogamy

We introduce three sets of basic axioms/properties a measure should satisfy. They describe (i) matching matrices that should be classified as equally assortative, (ii) matching matrices where one must be classified as more assortative than the other, and (iii) matching matrices that should be classified as most assortative. For generality, whenever possible, we define axioms that a preorder should satisfy, rather than an index. We say that a total preorder  $\succeq$  is *induced by* index I if for all matching matrices M and M',  $I(M) > I(M') \Leftrightarrow M \succ M'$ 

<sup>&</sup>lt;sup>9</sup>In general, the iso-assortative curves of a fixed matching matrix can be derived for any measure and its upper contour set can be used to compare and contrast any pairs of measures. In addition, when we relax the restriction of b = c, we can construct iso-assortative surfaces and upper contour sets of M in three-dimensional planes. We illustrate the conflicts between aggregate likelihood ratio and odds ratio in the three-dimensional space in Appendix Figure 2.

and  $I(M) = I(M') \Leftrightarrow M \sim M'$ . Two indices I and I' are *preorder-equivalent* if the total preorders  $\succeq_I$  and  $\succeq_{I'}$  they induce are equivalent. An index is said to satisfy an axiom if the total preorder induced by the index satisfies it.

### 3.1 Invariance axioms

We start with axioms that define two equally assortative matching matrices. First, when a matching matrix is scaled up or down by a positive constant without changing the relative composition, the assortativeness should be evaluated as unchanged.

**Scale Invariance.** For all M and 
$$\lambda > 0$$
,  $M \sim \lambda \cdot M$ .

Given scale invariance, we can transform M into a contingency table by dividing each cell by |M|. Next, the assortativeness should not be affected if we change the sides of the market. Effectively, we switch the masses of high-low (b) and low-high (c) pairs.

Side Invariance. The matching stays equally assortative when sides are switched:  $\frac{a \mid b}{c \mid d} \sim \frac{a \mid c}{b \mid d}.$ 

Neither should switching the types affect the assortativeness. Swapping the labels of the two types is essentially swapping high-high (a) and low-low (d) pairs and swapping high-low (b) and low-high (c) pairs.

**Type Invariance.** The matching stays equally assortative when types are switched:  $\frac{a \ b}{c \ d} \sim \frac{d \ c}{b \ a}.$  We refer to all three together as *invariance axioms*. These are basic axioms an appropriate measure should satisfy. While type-specific likelihood ratio fails type invariance, all aforementioned measures satisfy all three axioms of invariance (see summary in Table 1).

### 3.2 Monotonicity axioms

Next, we define notions of monotonicity that specify when one matching matrix is more assortative than another. We define two notions of monotonicity, one based on comparisons of matching matrices with the same marginal distributions and the other based on comparisons of matching matrices with different marginals.

We compare positive matching matrices (all entries are positive). They are the nonmaximally assortative matching matrices.

First, consider one notion of monotonicity that compares two matching matrices of the same marginal distributions. Matching matrices that share the same marginal distributions as (a, b, c, d) are of the form  $(a + \epsilon, b - \epsilon, c - \epsilon, d + \epsilon)$ —essentially, a one-parameter family.

Marginal Monotonicity. For any positive matching matrices M = (a, b, c, d) > 0and M' = (a', b', c', d') > 0 with the same marginal distributions (i.e., a + c = a' + c', a + b = a' + b', d + b = d' + b', and d + c = d' + c'),  $M \succ M'$  if and only if a > a'(equivalently, b < b', c < c', or d > d').

All aforementioned measures satisfy marginal monotonicity.

Next, we compare matching matrices that differ in marginal distributions, but differ in a minimal way: by one cell. For matching matrices that are not maximally assortative, when pairs of like types (i.e., the terms on the diagonal of the matrix) increase, the matching becomes more assortative.

**Diagonal Monotonicity.** For any positive matching matrix M > 0 and  $\epsilon > 0$ ,

$$\frac{a+\epsilon}{c} \xrightarrow{b} \succ \frac{a}{c} \xrightarrow{b} and \frac{a}{c} \xrightarrow{b} \frac{a}{c} \xrightarrow{b} \frac{a}{c} \xrightarrow{b} \frac{b}{c}$$

We analogously define *off-diagonal monotonicity*: When pairs of unlike types (i.e., the terms off the diagonal of the matrix) increase, the matching becomes less assortative.

**Off-Diagonal Monotonicity.** For any positive matching matrix M > 0 and  $\epsilon > 0$ ,  $\frac{a \ b}{c \ d} \succ \frac{a \ b + \epsilon}{c \ d} \text{ and } \frac{a \ b}{c \ d} \succ \frac{a \ b}{c + \epsilon \ d}.$ 

Claim 1. Diagonal and off-diagonal monotonicities imply marginal monotonicity.

**Proof of Claim 1.** Suppose positive matrices M = (a, b, c, d) and M' = (a', b', c', d') have the same marginal distributions, and suppose a > a'. Because they share the same marginals, a - a' = b' - b = c' - c = d - d' (derived from a + c = a' + c', a + b = a' + b', d + b = d' + b', and d + c = d' + c'). Hence, M' can be represented as

$$M' = \frac{a - (a - a')}{c + (c' - c)} \frac{b + (b' - b)}{d - (d - d')} = \frac{a - (a - a')}{c + (a - a')} \frac{b + (a - a')}{d - (a - a')}$$

Then,

$$M' = \frac{a - (a - a')}{c + (a - a')} \frac{b + (a - a')}{d - (a - a')} \prec \frac{a}{c + (a - a')} \frac{b + (a - a')}{d} \prec \frac{a}{c} \frac{b}{d} = M,$$

where the first  $\prec$  follows from diagonal monotonicity and the second  $\prec$  follows from offdiagonal monotonicity. Conversely, however, marginal monotonicity does not necessarily imply diagonal or offdiagonal monotonicity, because marginal monotonicity solely specifies relations for matchings with the same marginal distributions and by itself has no implications for markets with different marginal distributions.

Let us examine whether the aforementioned measures satisfy these monotonicity axioms. The type-specific likelihood ratio satisfies off-diagonal monotonicity but fails diagonal monotonicity:  $I_{L1}(1, 1, 1, 6) = 2.25 > I_{L1}(2, 1, 1, 6) \approx 2.22 > I_{L1}(6, 1, 1, 6) \approx 1.74$ . To investigate further, we can take the partial derivative of  $I_{L1}$  with respect to a. Diagonal monotonicity tends to fail when one of the cells is relatively small. The aggregate likelihood ratio satisfies off-diagonal monotonicity but fails diagonal monotonicity:

$$\frac{\partial I_L}{\partial a} = \frac{(a+d)(d-a)(b+c)}{\left[(a+b)(a+c) + (d+b)(d+c)\right]^2} < 0 \text{ when } a > d.$$

The other aforementioned measures satisfy diagonal and off-diagonal monotonicity.

### 3.3 Homogamy axioms

A natural requirement is that any maximally positive-assortative matching should be a maximal element for preorder  $\succeq$ . Following Robbins (2009), we call it the *maximum homogamy* property<sup>10</sup>

**Maximum Homogamy.** For any 
$$M$$
,  $ad > 0$ , and  $bc = 0$ :  $(a, b, c, d) \succeq M$ .

A less restrictive condition suggests that fully positive-assortative matching should be perceived as the most positive-assortative; we call this the *full homogamy* property.<sup>11</sup>

**Full Homogamy.** For any M and ad > 0:  $(a, 0, 0, d) \succeq M$ .

<sup>&</sup>lt;sup>10</sup>HMax in Chiappori, Costa Dias, and Meghir (2022).

<sup>&</sup>lt;sup>11</sup>Weak HMax in Chiappori, Costa Dias, and Meghir (2022).

Note that maximum homogamy is a stricter condition than full homogamy, because it requires not only matrices of the form (a, 0, 0, d) to be maximal elements, but also matrices of the form (a, 0, c, d) or (a, b, 0, d).

One can readily check that the odds ratio satisfies maximum homogamy, and hence full homogamy, since it becomes infinite when b = 0 or c = 0. By construction, we have  $I_{MD}(M) \leq 1$  for all M and  $I_{MD}(a, b, 0, d) = 1$ . Therefore minimum distance satisfies maximum homogamy. The correlation index obviously satisfies full homogamy, since the correlation is then equal to 1. However, it does not satisfy maximum homogamy. For instance,  $I_{Corr}(45, 5, 5, 45) = .8 > .41 = I_{Corr}(40, 40, 0, 20)$ .

Likelihood ratios fail full homogamy, and hence maximum homogamy. Compare matching matrices M = (5, 5, 5, 85) and M' = (50, 0, 0, 50) corresponding, for instance, to two different cohorts in the same economy. The distribution of education is independent of gender in both M and M', but the proportion of educated people has increased from 10% in M to 50% in M'. Cohort M exhibits PAM in the usual sense (more people on the diagonal than would obtain under random matching); yet, 50% of educated people marry an uneducated spouse. Cohort M' displays perfect sorting, with all educated individuals marrying together. The college-based likelihood ratio yields  $I_{L1}(M) = 5$  and  $I_{L1}(M') = 2$ : Assortativeness has decreased from M to M'.

### 3.4 Summary

Table 1 summarizes whether the measures satisfy the basic invariance, monotonicity, and homogamy axioms. The type-specific likelihood ratio does not satisfy type invariance, because relabeling the types would change the measure. The aggregate likelihood ratio fails diagonal monotonicity, maximum homogamy, and full homogamy. The correlation index fails maximum homogamy.

The odds ratio, normalized trace, and minimum distance satisfy all the basic axioms. That begs the question, what would be axioms that can uniquely characterize these mea-

	Invari	iance a	xioms	Mo	notonicity a	Homogamy axioms		
	Scale	Side	Type	Marginal	Diagonal	Off-diagonal	Maximum	Full
Odds ratio	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Type-specific likelihood ratio	$\checkmark$	$\checkmark$	Х	$\checkmark$	X	$\checkmark$	Х	Х
Aggregate likelihood ratio	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	Х	$\checkmark$	Х	Х
Normalized trace	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Minimum distance	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Correlation	<ul> <li>✓</li> </ul>	$\checkmark$	$\checkmark$	✓	$\checkmark$	$\checkmark$	X	$\checkmark$

Table 1: Do the measures satisfy the invariance, monotonicity, and homogamy axioms?

sures? For the odds ratio and the normalized trace, we answer this question next.

## 4 Characterization results

Table 1 shows that, except for the type-specific likelihood ratio, all the measures listed satisfy the invariance axioms and at least one notion of monotonicity: marginal monotonicity. Hence, the basic axioms are not sufficient to distinguish the measures and provide a definitive answer regarding what measure to use.

We need additional axioms to distinguish and characterize the different measures. We will introduce what we call *characterization axioms*, such that a measure will be the unique one that satisfies both the basic axioms and an additional axiom, which essentially highlights the special property of the measure.

### 4.1 Marginal independence and odds ratio

We first provide the characterization result regarding marginal independence (Edwards, 1963) and the odds ratio. A measure satisfies marginal independence if multiplying any row or any column of any matching matrix does not change the assortativeness order. Intuitively, two matching matrices differing in the number of (say) educated women will be equally assortative if educated women match to educated and uneducated men in the same proportion in both markets. This will happen, for instance, if there is an increase in the number of educated women in the market, and the sorting patterns of the additional educated women are the same as for the initial stock.

Marginal Independence (Edwards, 1963). For any M and  $\lambda > 0$ ,

$$\frac{a}{c} \frac{b}{d} \sim \frac{\lambda a}{c} \frac{\lambda b}{d} \sim \frac{a}{\lambda c} \frac{b}{\lambda d} \sim \frac{\lambda a}{\lambda c} \frac{b}{\lambda c} \sim \frac{a}{\lambda c} \frac{\lambda b}{\lambda c}.$$

Note that marginal independence is an ordinal property. Nonetheless, it is a strong condition. It implies scale, side, and type invariance. In addition,

Claim 2. Marginal independence and marginal monotonicity imply diagonal monotonicity and off-diagonal monotonicity.

In fact, more strongly, the three monotonicity properties are equivalent under marginal independence:

**Claim 3.** Suppose a measure satisfies marginal independence. It satisfies diagonal monotonicity if and only if it satisfies off-diagonal monotonicity if and only if it satisfies marginal monotonicity.

**Theorem 1.** The odds ratio induces the unique preorder that satisfies marginal monotonicity, marginal independence, and maximum homogamy.

Note that the sole role of maximum homogamy is to specify the maximum PAM to be the maximal element of the preorder.

It is interesting to refer to an older statistics literature that discusses the properties of measures of association in the case of paired attributes (i.e., in our case, husband's and wife's education). The property posed by Edwards (1963) states that the association should not be "influenced by the relative sizes of the marginal totals" (p. 110). That is, the measure should

not change if one starts from a matching M and doubles the mass of couples in which the man is educated (while keeping unchanged the ratio of educated versus uneducated wives of educated men).

Edwards (1963) justifies this property by posing that the measure must only be a function of the proportion of educated women whose husband is educated and the proportion of uneducated women whose husband is educated (and conversely), so that any population change that keeps these proportions constant should not affect the index. This imposes, in particular, that global changes in the marginal distributions of the characteristic on which people match, for instance a global increase in the mass of educated women, should not systematically impact the index; only changes in the odds of marrying different types of spouses should matter. The condition was later generalized by Altham (1970) to the  $n \times n$  case.

Among the indices just reviewed, only the odds ratio and its preorder-preserving transformations satisfy marginal independence. It is interesting to consider how the other indices violate this requirement. Consider matching matrices  $M_{\lambda} = (\lambda a, \lambda b, c, d)$  with ad > bc (PAM) and  $\lambda \geq 1$ . Suppose  $\lambda$  increases. The type- $\theta_1$  likelihood ratio *decreases* since  $\partial I_{L1}/\partial \lambda < 0$ , while the type- $\theta_2$  likelihood ratio *increases*. All other indices may increase or decrease, depending on parameters.

### 4.2 Population decomposability and normalized trace

The next set of characterization axioms will endow the measures with cardinal interpretations. Namely, the assortativeness of a matching matrix will be a weighted average or sum of the measures of matching matrices in the submarkets decomposed from the original market. Depending on the weights used, different decomposability axioms correspond to different measures.

Consider the following axiom in which the weight to average is the population size of submarkets.

**Population Decomposability.** For any positive matching M > 0 and M' > 0,

$$I(M + M') = \frac{|M|}{|M| + |M'|} I(M) + \frac{|M'|}{|M| + |M'|} I(M').$$
(3)

Equivalently, for any positive matchings  $M_n > 0$ , n = 1, 2, ..., N,

$$I\left(\sum_{n} M_{n}\right) = \sum_{n} \left[\frac{|M_{n}|}{\sum_{n}|M_{n}|}I(M_{n})\right].$$
(4)

Equivalent expressions (3) and (4) mean that for any decomposition of a market, the assortativeness index of the matching in the entire market is a weighted average of the same index in its decomposed submarkets.<sup>12</sup> Practical and relevant examples of the decomposition of marriage markets include (i) by geographic regions, (ii) by metropolitan status, such as urban and rural marriages, or (iii) by race, such as into marriages between whites, those between nonwhites, and interracial marriages between whites and nonwhites.

As explained above, marginal monotonicity and the invariance axioms cannot imply diagonal monotonicity or off-diagonal monotonicity. However,

# **Claim 4.** Marginal monotonicity and population decomposability, together with scale invariance and type invariance, imply diagonal monotonicity and off-diagonal monotonicity.

We show in the theorem below that population decomposability is a characterization axiom for normalized trace. That is, the normalized trace is the unique index, up to positive

$$\begin{split} I(M) &= \frac{|M_1|}{|M_1| + |M_2 + M_3|} I(M_1) + \frac{|M_2 + M_3|}{|M_1| + |M_2 + M_3|} I(M_2 + M_3) \\ &= \frac{|M_1|}{|M_1| + |M_2| + |M_3|} I(M_1) + \frac{|M_2 + M_3|}{|M_1| + |M_2| + |M_3|} \left[ \frac{|M_2|}{|M_2| + |M_3|} I(M_2) + \frac{|M_3|}{|M_2| + |M_3|} I(M_3) \right] \\ &= \frac{|M_1|}{|M_1| + |M_2| + |M_3|} I(M_1) + \frac{|M_2|}{|M_1| + |M_2| + |M_3|} I(M_2) + \frac{|M_3|}{|M_1| + |M_2| + |M_3|} I(M_3). \end{split}$$

<sup>&</sup>lt;sup>12</sup>Expressions (3) and (4) are equivalent for the following reasons. It is straightforward that expression (4) implies expression (3), because expression (4) when n = 2 is expression (3). Expression (3) implies expression (4) by induction; below is the step to show that (3) implies (4) for n = 3: For matching M decomposed into three positive matchings  $M_1$ ,  $M_2$ ,  $M_3$ ,

affine transformation, that satisfies population decomposability and the basic invariance and monotonicity axioms. Under population decomposability, Claims 1 and 4 combine to imply the equivalence of the two notions of monotonicity—diagonal and off-diagonal monotonicity and marginal monotonicity—and so the characterization result holds for both notions of monotonicity.

**Theorem 2.** An index satisfies the basic axioms (i.e., of invariance, monotonicity, and homogamy) and population decomposability if and only if it is a positive affine transformation of normalized trace.

Equivalently, the normalized trace is the unique index, up to positive affine transformation, that satisfies the three invariance axioms, marginal monotonicity, maximum homogamy, and population decomposability. Or alternatively, it is the unique index, up to positive affine transformation, that satisfies the three invariance axioms, diagonal monotonicity, off-diagonal monotonicity, maximum homogamy, and population decomposability.

In addition, because of uniqueness up to a positive affine transformation, the index satisfying all basic axioms and population decomposability is uniquely determined if the range is specified. For example, when the range is [0, 1], the unique index is the normalized trace. Maximum homogamy is used to pin down the boundary cases of maximally but not fully positive-assortative matchings, i.e., (a, b, c, d) when b = 0 or c = 0 (but not both). We restricted the normalized trace of a matching of form (a, b, 0, d), where abd > 0, to be 1. If we drop maximum homogamy as a criterion, then the normalized trace without this restriction namely, I(a, b, c, d) = (a+d)/(a+b+c+d) for any matching matrix (a, b, c, d)—also satisfies the other aforementioned properties.

### 4.3 Random decomposability and aggregate likelihood ratio

The aggregate likelihood ratio is characterized by a decomposability axiom in which the summation weights are the expected mass of pairs of like types under random matching.

**Random Decomposability.** For any positive matching matrices M and M',

$$I(M + M') = \frac{r(M)}{r(M + M')}I(M) + \frac{r(M')}{r(M + M')}I(M'),$$

where r(M) indicates the expected mass of pairs of like types under random matching in market M:

$$r(M) \equiv \frac{a+b}{|M|} \frac{a+c}{|M|} |M| + \frac{d+b}{|M|} \frac{d+c}{|M|} |M| = \frac{(a+b)(a+c) + (d+b)(d+c)}{a+b+c+d}.$$

Equivalently, for any positive  $M_n$ , n = 1, 2, ..., N,

$$I\left(\sum_{n} M_{n}\right) = \sum_{n} \left[\frac{r(M_{n})}{r\left(\sum_{n} M_{n}\right)}I(M_{n})\right]$$

**Theorem 3.** An index satisfies the three invariance axioms, marginal monotonicity, and random decomposability if and only if it is a positive multiple of aggregate likelihood ratio.

We make two comments. First, note that the aggregate likelihood ratio does not satisfy maximum homogamy, diagonal monotonicity, or off-diagonal monotonicity, so we do not have a characterization result that connects those axioms and the aggregate likelihood ratio. Second, note that the class of measures that satisfy the axioms in Theorem 3 must be a positive multiple—i.e., a positive affine transformation without an intercept—of the aggregate likelihood ratio. This is because the weights r(M)/r(M + M') and r(M')/r(M + M') do not necessarily add up to be 1. Nonetheless, this characterization result gives the aggregate likelihood ratio a cardinal interpretation.

To address the observation that the weights do not necessarily add up to one, consider an axiom that involves weighted averages that add up to one and compares markets with proportional marginal distributions. Marginal Random Decomposability. For all M > 0 and M' > 0 such that M/|M| and M'/|M'| have the same marginal distributions,

$$I(M + M') = \frac{r(M)}{r(M) + r(M')}I(M) + \frac{r(M')}{r(M) + r(M')}I(M').$$

Aggregate likelihood ratio satisfies this axiom, but it is not the unique measure that does so. For example, the normalized trace—and as a result, any linear combination of aggregate likelihood ratio and normalized trace—also satisfies this axiom.

## 5 Structural interpretation of the odds ratio

It is important to note that among the various indices, the odds ratio has a known structural interpretation. Specifically, assume that the observed matching behavior constitutes the stable equilibrium of a frictionless matching model under transferable utility. Assume, furthermore, that the surplus generated by a match between woman x belonging to category  $\theta_i$  and man y belonging to category  $\theta_j$  takes the separable form

$$s(x,y) = Z^{\theta_i \theta_j} + \alpha_x^{\theta_j} + \beta_y^{\theta_i}, \tag{5}$$

where Z is a deterministic component depending only on individual educations and the  $\alpha$ 's and  $\beta$ 's are random shocks reflecting unobserved heterogeneity among individuals.<sup>13</sup> It is now well known (Graham, 2011; Chiappori, 2017) that, keeping constant the distribution of the shocks, assortativeness is related to the supermodularity of the matrix  $Z^{\theta_i\theta_j}$ , i.e., in the two-type case, to the sign of the supermodular core  $Z^{\theta_1\theta_1} + Z^{\theta_2\theta_2} - Z^{\theta_1\theta_2} - Z^{\theta_2\theta_1}$ . More importantly, if, following the seminal contribution by Choo and Siow (2006), one assumes that the random shocks follow type-1 extreme value distributions—the so-called *Separable Extreme Value* model—then the supermodular core equals twice the odds ratio. Hence, the

<sup>&</sup>lt;sup>13</sup>Transferable-utility models satisfying condition (5) are referred to as separable models in the literature.

odds ratio represents the average gain per couple from positive-assortative matching over negative-assortative matching.

This structural interpretation is especially useful for disentangling possible changes in the value of different matches from the mechanical effect of variations in the marginal distributions of education among individuals. "Structural changes," here, can only affect either the matrix Z or the distributions of random shocks. It is also useful for constructing counterfactual simulations, since the same structure can be applied to different distributions of education by genders, using standard techniques to solve for the stable equilibrium of the corresponding matching game.<sup>14</sup>

We make two remarks at this point. First, as clearly pointed out by Choo and Siow (2006) as well as by the subsequent literature (Galichon and Salanié, 2022; Chiappori and Salanié, 2016), this structural model can be identified (in the econometric sense) from matching patterns, but only under strong parametric restrictions. For instance, the initial Choo and Siow (2006) framework, which generates the odds ratio as an estimator of the supermodularity of the surplus, is *exactly* identified under the assumption that the random shocks  $\alpha$ 's and  $\beta$ 's follow type-1 extreme value distributions. Indeed, the consensus is that, to be identified from the sole observation of matching patterns, any structural model would require strong parametric assumptions regarding the distribution of random preference shocks. It follows that, in general, the ranking (in terms of assortativeness, i.e., supermodularity of the deterministic surplus) may vary with the specific assumptions made on the distribution of the stochastic factors.

Several routes can be followed to overcome this limitation. Following Chiappori, Salanié, and Weiss (2017), one may consider repeated cross sections and impose restrictions on how the structural components change over time. Alternatively, Chiappori, Costa Dias, and Meghir (2018) argue that direct observation of post-marital behavior provides additional information on the surplus (since the latter, under transferable utility, is simply the sum

<sup>&</sup>lt;sup>14</sup>See Chiappori et al. (2020) and Chiappori, Costa Dias, and Meghir (2020) for applications.

of individual utilities and can be estimated from the observation of the demand functions); this information can be used in particular to relax parametric assumptions made on the stochastic components of the model. In a recent contribution, Gualdani and Sinha (2023) also show how partial (i.e., set) identification may obtain using general assumptions on the distribution of stochastic shocks (for instance, independence of taste shocks from covariates and quantile, or symmetry restrictions).

Second and more importantly, while robust examples can be given where an assortativeness ranking is reversed when the assumptions regarding stochastic distributions are changed, one can nevertheless define conditions under which the ranking will be the same for any separable transferable utility model, *irrespective of the stochastic distribution*, provided the latter satisfy some basic properties (such as independence). Specifically, Chiappori, Costa Dias, and Meghir (2020) show that if two PAM M and M' are such that:

$$\frac{a}{a+b} \ge \frac{a'}{a'+b'}, \ \frac{a}{a+c} \ge \frac{a'}{a'+c'}, \ \frac{d}{b+d} \ge \frac{d'}{b'+d'} \ \text{and} \ \frac{d}{c+d} \ge \frac{d'}{c'+d'}, \tag{6}$$

then irrespective of the stochastic distributions of  $\alpha$  and  $\beta$  (provided they are independent from each other and from the observed characteristics), the deterministic surplus corresponding to M will be more supermodular than M'; this is the *Generalized Separable (GS)* criterion in Chiappori, Costa Dias, and Meghir (2020). As a result, for any stochastic distributions of  $\alpha$ 's and  $\beta$ 's, while the numerical value of the supermodular cores will depend on the choice of distributions, the structural model will always rank M above M' in terms of assortativeness.

In other words, one can define a preorder that is totally robust to changes in distributional assumptions. The price to pay for this generalization, however, is that the preorder is no longer complete: If some inequalities in (6) are satisfied while others are violated, the matching matrices simply cannot be compared.

## 6 Multiple types

There are 14 different education categories in the census, for example. It is worthwhile to consider how to compare the assortativeness of matching matrices of more than two types (if we must compare). Consider matching  $M = (\mu_{ij})_{i,j \in \{1,\dots,n\}}$  for  $n \ge 3$ .

### 6.1 Extensions from two types

The aggregate likelihood ratio and normalized trace can be naturally extended to multi-type markets, but the odds ratio does not have a natural extension. The *normalized trace for multi-type markets* is

$$I_{tr}(M) = \begin{cases} 1 & \text{if } \mu_{ij}\mu_{ji} = 0 \ \forall i \text{ and } j \neq i \\ 0 & \text{if } \operatorname{tr}(M) \equiv \sum_{i=1}^{n} \mu_{ii} = 0 \\ \operatorname{tr}(M)/|M| & \text{otherwise.} \end{cases}$$

The aggregate likelihood ratio for multi-type markets is

$$I_L(M) \equiv \frac{tr(M)/|M|}{\sum_{i=1}^n \left(\frac{\sum_{j=1}^n \mu_{ij}}{|M|}\right) \left(\frac{\sum_{j=1}^n \mu_{ji}}{|M|}\right)} = \frac{|M|(\sum_{i=1}^n \mu_{ii})}{\sum_{i=1}^n (\sum_{j=1}^n \mu_{ij})(\sum_{j=1}^n \mu_{ji})}.$$

Theorems 2 and 3 can be naturally extended so that normalized trace and the aggregate likelihood ratio are the unique indices that satisfy the sets of axioms (appropriately modified) in those theorems. Hence, normalized trace and the aggregate likelihood ratio can still be used to measure the assortativeness of multi-type markets. In addition, we can extend the measures to measure assortativeness in (i) markets with unmarried and (ii) one-sided (e.g., homosexual) matching markets. See Zhang (2024) for more detailed theoretical discussions and empirical analyses of these markets.

### 6.2 Robustness to categorization

Consider three types  $\theta_i$ ,  $\theta_j$ , and  $\theta_k$ . When two types  $\theta_i$  and  $\theta_j$  merge so that the three categories are partitioned to  $\pi = \{\{\theta_i, \theta_j\}, \{\theta_k\}\}\}$ , the matching given this categorization becomes

$$M^{\pi} = \begin{pmatrix} \mu_{ii} + \mu_{ij} + \mu_{ji} + \mu_{jj} & \mu_{ik} + \mu_{jk} \\ \mu_{ki} + \mu_{kj} & \mu_{kk} \end{pmatrix}$$

We say that  $M_1$  is more assortative than  $M_2$  if and only if  $M_1^{\pi}$  is more assortative than  $M_2^{\pi}$  for all partitions  $\pi$ . For example, with normalized trace, for M and M' such that |M| = |M'| and  $\operatorname{tr}(M) = \operatorname{tr}(M')$ , M is more assortative than M' if and only if  $\mu_{12} + \mu_{21} \ge \mu'_{12} + \mu'_{21}$ ,  $\mu_{13} + \mu_{31} \ge \mu'_{13} + \mu'_{31}$ , and  $\mu_{23} + \mu_{32} \ge \mu'_{23} + \mu'_{32}$ . Formally, we define *robustness to categorization* as follows.

**Robustness to Categorization.** Let categorization  $\Pi$  denote the set of partitions of types to be considered. Let  $M^{\pi}$  denote the matching under partition  $\pi \in \Pi$ . A preorder  $\succeq$  is robust to categorization  $\Pi$  when for all matrices  $M_1$  and  $M_2$ ,  $M_1 \succeq M_2$  if and only if  $M_1^{\pi} \succeq M_2^{\pi}$  for any partition  $\pi \in \Pi$ , and  $M_1 \succ M_2$  if and only if  $M_1^{\pi} \succ M_2^{\pi}$ for any partition  $\pi \in \Pi$ .

**Theorem 4.** Let  $\Pi$  be a nondegenerate categorization. No total preorder satisfies marginal monotonicity and robustness to categorization  $\Pi$ .

A counterexample suffices for the claim. Although the counterexample is one of threetype matching matrices, it suffices for the claim to be valid for matching matrices with more than three types. **Proof of Theorem 4.** Consider matrices M and  $M_{\epsilon}$  of three types  $\theta_1$ ,  $\theta_2$ , and  $\theta_3$ ,

Under partition  $\pi_{12} = \{\{\theta_1, \theta_2\}, \{\theta_3\}\}$ , we group  $\theta_1$  and  $\theta_2$ ,

$$M^{\pi_{12}} = \frac{4/9}{2/9} \frac{2/9}{1/9} \text{ and } M^{\pi_{12}}_{\epsilon} = \frac{4/9 + \epsilon}{2/9 - \epsilon} \frac{2/9 - \epsilon}{1/9 + \epsilon} = M^{\pi_{12}} + \frac{+\epsilon}{-\epsilon} \frac{-\epsilon}{-\epsilon}$$

Under partition  $\pi_{23} = \{\theta_1, \{\theta_2, \theta_3\}\}$ , we group  $\theta_2$  and  $\theta_3$ ,

$$M^{\pi_{23}} = \frac{1/9}{2/9} \frac{2/9}{4/9} \text{ and } M^{\pi_{23}}_{\epsilon} = \frac{1/9 - \epsilon}{2/9 + \epsilon} \frac{2/9 + \epsilon}{4/9 - \epsilon} = M^{\pi_{23}} + \frac{-\epsilon}{+\epsilon} \frac{+\epsilon}{+\epsilon}$$

By marginal monotonicity,

$$M^{\pi_{12}} \prec M^{\pi_{12}}_{\epsilon}$$
 and  $M^{\pi_{23}} \succ M^{\pi_{23}}_{\epsilon}$ .

Hence, there does not exist a total preorder that satisfies marginal monotonicity and robustness to categorization.  $\hfill \Box$ 

The impossibility result suggests that we must resort to partial rankings to satisfy some notion of monotonicity and robustness to categorization. Existing partial rankings such as supermodular stochastic order (Meyer and Strulovici, 2012, 2015) and positive quadrant dependence order (Anderson and Smith, 2024) are natural candidates. However, they only compare matching matrices of the same marginals. Hence, further investigations to compare matching matrices of differential marginal distributions are needed.

## 7 Educational homogamy in the US

Our paper was in part motivated by the conflicting results on how educational homogamy has changed in the US. We now consider its evolution for recent cohorts, through the lens of our discussion. Our goal is to compare and contrast the answers given by various indices to the same basic question: Did educational homogamy increase between different cohorts? Gihleb and Lang (2020) consider this issue, using a variety of indices (correlation, rank correlation, Goodman and Kruskal's  $\gamma$ , Kendall's  $\tau$  adjusted and not adjusted for ties). A difference from our approach is that they consider changes affecting the entire distribution of education and many classes, whereas we in addition consider how assortativeness changed locally at the top of the education distribution using our  $2 \times 2$  framework.

We use the March extract of the US Current Population Survey, from where we select the subsample of married individuals (including legal marriage and cohabitation) observed when aged 35-44 and belonging to the 1950s or the 1970s cohorts.<sup>15</sup> This is an interesting comparison to make because just after the 50's cohort educational attainment of women accelerated and overtook that of men, which also increased but less so.(Chiappori, Iyigun, and Weiss, 2009; Zhang, 2021). Relatedly, the returns to investments in human capital increased substantially during the 1980s and later (Katz and Murphy, 1992). Educational and marital choices by the older 1950s generation were mostly made before that period; in contrast, individuals born in the 1970s, when choosing both their education level and their spouse, were fully aware of the new context. Consequently, it has been argued that homogamy should have increased between these two cohorts (for instance, Chiappori, 2017; Chiappori, Salanié, and Weiss, 2017), which is what we reconsider here. Specifically, we define the 1950s cohort as married couples where at least one partner was born in the 1950-59 period and similarly for the 1970s cohort.

Following Gihleb and Lang (2020), who highlight the sensitivity of empirical changes

 $<sup>^{15}\</sup>mathrm{In}$  Appendix Table 5 we present comparisons of the 70's with other cohorts, which also show interesting patterns.

in assortativeness to separating college graduates and post-graduates over this period, we consider 5 education groups: high-school dropouts (HSD), high-school graduates (HS), some college education but no degree (SC), 4-year college degree (C) and post-graduate degree (meaning more than bachelor's degree, PG).

Category	HSD	HS	$\mathbf{SC}$	С	$\mathbf{PG}$	Marg Dist: Men			
1950s Cohort									
Men		W	Vomer	ı					
HSD	4.7	3.9	1.1	0.2	0.0	9.9			
HS	3.1	20.0	7.1	2.4	0.6	33.2			
$\mathbf{SC}$	0.9	9.1	10.8	3.8	1.0	25.6			
С	0.2	3.9	5.8	7.8	2.0	19.6			
PG	0.1	1.1	2.4	4.4	3.7	11.6			
Marg Dist: Women	8.9	37.9	27.2	18.6	7.4				
1970s Cohort Men Women									
HSD	4.5	2.4	1.2	0.4	0.1	8.6			
HS	2.0	13.5	7.8	3.8	1.3	28.4			
$\mathbf{SC}$	0.6	4.9	11.5	5.9	2.4	25.3			
С	0.2	20	18	116	4.0	92.4			

Table 2: Marginal distributions of education and matching in the 1950s and the 1970s cohorts

SC	0.6	4.9	11.5	5.9	2.4	25.3
$\mathbf{C}$	0.2	2.0	4.8	11.6	4.9	23.4
$\mathbf{PG}$	0.1	0.5	1.8	5.3	6.6	14.3
Marg Dist: Women	7.3	23.4	27.0	27.0	15.3	

Educational categories: HSD: High School dropout, HS: High School, SC: Some College, C: College, PG: Post-graduate. Numbers represent cell percentages.

Table 2 shows the marginal distributions of education for men and women and the resulting matching patterns. Both panels show evidence of positive sorting, with most of the mass concentrated on the main diagonal. They also show how women with college or more education increased dramatically between these two cohorts, making this group larger than the corresponding one for men, which also increased but not by nearly as much. These changes in the marginal distributions are at the heart of the contradictory results of the various measures. The question is whether sorting became more positive between the two cohorts.

Table 3 presents the results. For each index and each matching (sub)matrix, we present its change across the two cohorts, with a positive number indicating an increase in positive sorting. Since we are testing differences across many indices and many matching matrices, we adjust our *p*-values for multiple testing using the Romano-Wolf stepdown approach (Romano and Wolf, 2005). The table presents two sets of *p*-values: in round brackets are the most conservative, allowing for all 30 estimates; in square brackets are *p*-values adjusted for multiple testing within each submarket only.

Table 3: Marital assortativeness – comparing cohorts born between 1950-59 and 1970-75

		Si	ubmarkets	Global indices		
		C&PG vs SC	PG vs C	C vs SC	C&PG vs others	5 educ levels
		(1)	(2)	(3)	(4)	(5)
(1)	Odds ratio	1.173 (.00) [.00]	-0.400 (.04) [.01]	0.920 (.00) [.00]	0.639 (.11) [.02]	
(2)	L1 ratio (high ed)	-0.046 (.00) [.00]	-0.074 (.00) [.00]	-0.025 (.10) [.02]	-0.421 (.00) [.00]	
(3)	L2 ratio (low ed)	0.246 (.00) [.00]	0.011 (.32) [.21]	0.113 (.00) $[.00]$	0.155 (.00) $[.00]$	
(4)	Weighted L ratio	-0.013 (.17) [.09]	-0.003 (.95) [.78]	0.058 (.00) $[.00]$	0.149 (.00) [.00]	0.223 (.00) [.00]
(5)	Normalized trace	0.041 (.00) [.00]	-0.004 $(.76)$ $[.55]$	0.024 (.00) [.00]	-0.017 (.00) [.00]	0.007 (.07) [.01]
(6)	$\chi^2$	0.029 (.00) [.00]	-0.008 (.31) [.20]	0.035 (.00) $[.00]$	0.040 (.00) [.00]	
(7)	Minimum distance	0.001 (.88) [.88]	-0.099 (.00) [.00]	0.028 (.04) [.01]	0.026 (.00) [.00]	

Notes: Education attainment in 5 groups: post-graduate degrees (PG, more than 4-years college), 4-years college degrees (C), some college education but no degree (SC), high school qualifications (HS), and no qualifications (HSD). First column refers to matching matrices between graduates (graduates and post-graduates taken together) and those with some college education but not a degree; column 2: post-graduates and 4-year college graduates; column 3: college graduates and some college; column 4: college graduates and everyone else together; column 5: all 5 education groups separate. For each index, the top row shows estimates of the difference in the respective index between the latest and earliest cohort; the number in round brackets shows p-values for 2-sided significance testing adjusted for multiple hypothesis using the stepdown method jointly for all the 30 measures on the table (Romano and Wolf, 2005; Romano, Shaikh, and Wolf, 2008; Romano and Wolf, 2016); the number in square brackets shows stepdown p-values but for each market (column) separately. *Data source*: March extract of the US Current Population Survey, subsample of legally married and cohabiting individuals observed when aged 35-44 and born in 1950-59 and 1970-75. Looking first at estimates in column 5 for the whole disaggregated  $5 \times 5$  sorting matrix, both the normalized trace and the aggregate likelihood ratio (the two indices that are designed for such a comparison) imply an overall increase in PAM, highly significant in the case of the likelihood ratio but only marginally so at conventional significance levels for the normalized trace. The change in the normalized trace is also close to zero, pointing to an increase of 0.7 percentage points in the mass of homogamous marriages across the two cohorts.<sup>16</sup> On the basis of these two measures, sorting increased or remained the same between these two cohorts.

Column 4 also displays estimates for the entire population of couples, but aggregating education in two larger groups: college-educated (graduates and post-graduates), and all those who did not gain college qualifications, so that the comparison is between two  $2 \times 2$  matrices. Under this aggregation, the aggregate likelihood ratio still shows a strongly significant increase in assortativeness, while the normalized trace shows a strongly significant drop. The latter confirms the conclusion in Gihleb and Lang (2020) that aggregation matters. Taken together, even measures satisfying the basic axioms (of invariance, monotonicity, homogamy) may point in opposite directions when comparing assortativeness in two matching matrices. In some instances, the ambiguity can be resolved by requiring that measures satisfy some characterization axiom, such as marginal independence, which leaves the odds ratio as the uniquely acceptable index. However, the impossibility result in Theorem 4 points to the limitations.

Consider now the likelihood ratios. In all columns 1-4 involving comparisons of  $2 \times 2$  matrices, the two type-specific likelihood ratio indices differ by whether the basis for its construction is the matching of the high education types with each other (L1 in row 2, type  $\theta_1$ ) or the low education types with each other (L2 in row 3, type  $\theta_2$ ). These conflicting findings

<sup>&</sup>lt;sup>16</sup>Note that the different indices vary in scale, so comparing the size of their changes is not meaningful. Most indices, including the odds ratio, lack a cardinal interpretation, and hence comparing how much they change across different markets is also not meaningful. However, the normalized trace stands out for having a cardinal interpretation, implied by its population decomposability property, and hence the size of changes in this measure across markets can be interpreted and compared.

between the two type-specific likelihood ratios are possible because they do not satisfy the basic axiom of type invariance. The aggregate likelihood ratio is a weighted average of the two, and depending on the relative mass associated with each, it can sometimes be positive or negative. Moreover, none of the likelihood ratios satisfies diagonal monotonicity or the homogamy axioms. This implies that in cases where there is a natural ranking of assortativeness revealed unambiguously by all other indices we examined (such as when one of the markets is fully homogamous, or when two markets differ by the mass in a single cell), it is possible that the likelihood ratios will show contrary findings. More generally, failure to satisfy these properties can result in the likelihood ratios frequently finding changes in assortativeness in the opposite direction of other measures that do satisfy them.

Submarket comparisons. Assortativeness can change differently, even in opposite directions, for different parts of the education distribution, making the overall indices difficult to interpret. Local differences in how much PAM changes across two matching matrices may also contribute to ambiguity in findings related to whole markets, as different measures weigh the various margins differently. Given the stark differences in university graduation rates across the two cohorts, we now focus on sorting at the top of the education distribution, between college-educated and those who attend for some time but drop out of college without a degree.

Column 1 considers the submarket of those with some college degree (C or PG) to those with some college attendance, but no 4-year degree (SC). Setting aside the likelihood ratios, all indices in column 1 show either a significant increase in sorting or no (significant) change at all (minimum distance). As we noted above, the direction of the type-specific likelihood ratios depends on how it is computed (L1 or L2), while the aggregate likelihood ratio points to a reduction in assortativeness but is not statistically significant at conventional levels.

To gain further insight, we separate those with a college degree by whether or not they obtained post-graduate degrees. Columns 2 and 3 show results for PG versus C and for

C versus SC respectively. While the indices consistently point to a significant increase in assortativeness between college graduates and some college (column 3), assortativeness may actually have decreased among graduates and post-graduates (column 2), though estimates are only statistically significant for the odds ratio and the minimum distance. The opposing changes in these two submarkets can help explain the ambiguous results in the literature, which most refer to as global measures of assortativeness.

	Si	Global index		
	C&PG vs SC	$\mathrm{PG}~\mathrm{vs}~\mathrm{C}$	C vs SC $$	C&PG vs others
	(1)	(2)	(3)	(4)
(1) $\Delta \frac{a}{a+b}$	0.126	-0.097	-0.077	0.181
	(.00) [.00]	(.00) [.00]	(.00) [.00]	(.00) [.00]
(2) $\Delta \frac{a}{a+c}$	-0.013	0.045	0.056	-0.018
	(.00) [.00]	(.00) $[.00]$	(.00) $[.00]$	(.00) [.00]
(3) $\Delta \frac{d}{d+b}$	0.069	-0.078	-0.010	0.020
	(.00) [.00]	(.00) [.00]	(.14) [.14]	(.00) $[.00]$
(4) $\Delta \frac{d}{d+c}$	-0.109	0.097	0.133	-0.105
	(.00) [.00]	(.00) $[.00]$	(.00) $[.00]$	(.00) [.00]

Table 4: Testing for assortativeness based on the structural model with no distributional assumptions, cohorts 1970s versus 1950s

Notes:  $\Delta$  signifies change from the 1950s to the 1970s cohort. Education attainment separates post-graduate degrees (PG more than 4-years college), 4-years college degrees (C), some college education but no degree (SC), and all others (high school qualifications HS, and no qualifications HSD). For each index, the top row shows estimates of the difference in the respective index between the latest and earliest cohort; the number in round brackets shows *p*-values for 2-sided significance testing adjusted for multiple hypothesis using the stepdown method jointly for all the 16 measures on the table (Romano and Wolf, 2005; Romano, Shaikh, and Wolf, 2008; Romano and Wolf, 2016); the number in square brackets shows stepdown *p*-values but for each market (column) separately. *Data source*: March extract of the US Current Population Survey, subsample of legally married and cohabiting individuals observed when aged 35-44 and born in 1950-59 and 1970-75.

#### Identifying changes in sorting based on a distribution-free structural approach.

We can apply our generalized separable conditions (6) to examine whether sorting increased or not. Table 4 shows estimates of the differences in the respective ratios. Identifying unambiguous changes in assortativeness across two markets requires that all differences in a column have the same sign: positive for an increase in PAM, and negative for a reduction. As shown in the table, the conditions for increased sorting under this criterion are not all simultaneously satisfied for any of the markets considered. This implies that, for the markets studied here, one can always find specific distributions for random preference terms in the separable transferable utility model such that the ranking would be reversed, a conclusion similar to that of Gualdani and Sinha (2023). While imposing marginal independence is attractive, we note that the odds ratio is not decomposable, while the normalized trace is (but does not satisfy marginal independence).

**Discussion.** Setting aside the likelihood ratios, all other indices point to an increase in assortativeness at the top of the education distribution with the exception of the very top, in the submarket involving those with post-graduate degrees matching with those with just 4-year college (bachelor's degree). The single exception is that of the normalized trace, which shows a drop in assortativeness when the entire market is aggregated in two big classes, contradicting its own predictions when assortativeness at the top is investigated at a more local level.

In other settings, however, there could be more ambiguity than we observe here, even between indices that satisfy all the basic axioms. In such cases, it will be especially important to consider the additional characterization properties that could help resolve the ambiguity. When comparing the markets for the two cohorts (1950s and 1970s), large changes in the marginal distributions stand out, especially for women for whom college attainment increased substantially. In cases such as this one, imposing marginal independence seems especially relevant: Changes in assortativeness relate to how the additional 1970s college women match with high versus low educated men in proportions similar or different to those for the college women in the 1950s. Imposing marginal independence means adopting the odds ratio, which under the Separable Extreme Value model admits an economic interpretation based on optimal partner choice and marriage market equilibrium. However, a question that remains open and in need of further investigation is the characterization of the complete class of structural economic models that satisfy the basic axioms and marginal independence. Discovering this, which is beyond the scope of this paper, will provide the complete link between the axiomatic approach and one based on optimizing economic behavior and the equilibrium in the marriage market.

## 8 Concluding Remarks

It is relatively simple to estimate whether there is positive assortative matching in a stochastic marriage market along the dimensions of a characteristic such as education. However, measuring the extent to which such assortative matching differs between two economies or between two points in time for the same economy is challenging when the marginal distributions of the characteristics also change.

In this paper, we show that different measures may generate different conclusions regarding the evolution of educational homogamy over time. We first take an axiomatic approach to better understand the underlying properties of commonly used measures. Namely, we axiomatize the odds ratio, highlighting the marginal independence property. We discuss a structural interpretation of the odds ratio. In addition, we also discuss the difficulties of comparing assortativeness for markets with more than two types. We then revisit the changes in assortative matching in the US between the 50's and the 70's cohort, between which there were large changes in the educational attainment of men and even more so for women. The odds ratio points to an increase in sorting with the exception of the very top of the education distribution that involves post-graduate degrees.

## References

- Abbott, Brant, Giovanni Gallipoli, Costas Meghir, and Giovanni L. Violante. 2019. "Education Policy and Intergenerational Transfers in Equilibrium." *Journal of Political Economy* 127 (6):2569–2624.
- Almar, Frederik, Benjamin Friedrich, Ana Reynoso, Bastian Schulz, and Rune Vejlin. 2023."Marital Sorting and Inequality: How Educational Categorization Matters." Mimeo.
- Altham, Patricia M. E. 1970. "The Measurement of Association of Rows and Columns for an r×s Contingency Table." Journal of the Royal Statistical Society. Series B (Methodological) 32:63–73.
- Anderson, Axel and Lones Smith. 2024. "The Comparative Statics of Sorting." American Economic Review 114 (3):709–751.
- Binder, Ariel J., Caroline Walker, Jonathan Eggleston, and Marta Murray-Close. 2023. "Race, Class, and Mobility in U.S. Marriage Market." Mimeo.
- Bouchet-Valat, M. 2014. "Changes in Educational, Social Class and Social Class of Origin Homogamy in France (1969–2011): Greater Openness Overall but Increased Closure of Elites." Revue Française de Sociologie (English Edition) 55:324–364.
- Cheremukhin, Anton A., Paulina Restrepo-Echavarria, and Antonella Tutino. 2024. "Marriage Market Sorting in the U.S." Mimeo.
- Chiappori, Pierre-André. 2017. Matching with Transfers: The Economics of Love and Marriage. Princeton University Press.
- Chiappori, Pierre-André, Monica Costa Dias, Sam Crossman, and Costas Meghir. 2020.
  "Changes in Assortative Matching and Inequality in Income: Evidence for the UK." *Fiscal Studies* 41:39–63.
- Chiappori, Pierre-André, Monica Costa Dias, and Costas Meghir. 2018. "The Marriage Market, Labor Supply, and Education Choice." Journal of Political Economy 126 (S1):S26– S72.

———. 2020. "Changes in Assortative Matching: Theory and Evidence for the US." NBER Working Paper 26932.

——. 2022. "Measuring Assortativeness in Marriage." Mimeo.

- Chiappori, Pierre-André, Murat Iyigun, and Yoram Weiss. 2009. "Investment in Schooling and the Marriage Market." *American Economic Review* 99 (5):1689–1713.
- Chiappori, Pierre-André and Bernard Salanié. 2016. "The Econometrics of Matching Models." *Journal of Economic Literature* 54:832–61.
- Chiappori, Pierre-André, Bernard Salanié, and Yoram Weiss. 2017. "Partner Choice, Investment in Children, and the Marital College Premium." American Economic Review 107:175–201.
- Choo, Eugene and Aloysius Siow. 2006. "Who Marries Whom and Why." Journal of Political Economy 114 (1):175–201.
- Ciscato, Edoardo, Alfred Galichon, and Marion Goussé. 2020. "Like Attract Like? A Structural Comparison of Homogamy Across Same-Sex and Different-Sex Households." Journal of Political Economy 128 (2):740–781.
- Ciscato, Edoardo and Simon Weber. 2020. "The Role of Evolving Marital Preferences in Growing Income Inequality." *Journal of Population Economics* 33:333–347.
- Edwards, A. W. F. 1963. "The Measure of Association in a 2×2 Table." Journal of the Royal Statistical Society A 126:109–114.
- Eika, Lasse, Magne Mogstad, and Basit Zafar. 2019. "Educational Assortative Mating and Household Income Inequality." Journal of Political Economy 127 (6):2795–2835.
- Fernández, Raquel and Richard Rogerson. 2001. "Sorting and Long-Run Inequality." Quarterly Journal of Economics 116 (4):1305–1341.
- Galichon, Alfred and Bernard Salanié. 2022. "Cupid's Invisible Hand: Social Surplus and Identification in Matching Models." *Review of Economic Studies* 89 (5):2600–2629.
- Gihleb, Rania and Kevin Lang. 2020. "Educational Homogamy and Assortative Mating Have Not Increased." *Research in Labor Economics* 48:1–26.

- Graham, Bryan. 2011. "Econometric Methods for the Analysis of Assignment Problems in the Presence of Complementarity and Social Spillovers." Handbook of Social Economics 1B (Benhabib, Jacksons and Bisin, Eds.) Amsterdam: North-Holland :965–1052.
- Greenwood, Jeremy, Nezih Guner, and John Knowles. 2003. "More on Marriage, Fertility, and the Distribution of Income." *International Economic Review* 44:827–862.
- Greenwood, Jeremy, Nezih Guner, Georgi Kocharkov, and Cezar Santos. 2014. "Marry Your Like: Assortative Mating and Income Inequality." American Economic Review: Papers and Proceedings 104:348–353.
- Gualdani, Cristina and Shruti Sinha. 2023. "Partial identification in matching models for the marriage market." *Journal of Political Economy* 131 (5):1109–1171.
- Hou, Yao, Ke Tang, Jingyuan Wang, Danxia Xie, and Hanzhe Zhang. 2022. "Assortative Mating on Blood Type: Evidence from One Million Chinese Pregnancies." Proceedings of the National Academy of Sciences 119 (51):e2209643119.
- Katz, Lawrence F. and Kevin M. Murphy. 1992. "Changes in Relative Wages, 1963–1987: Supply and Demand Factors." Quarterly Journal of Economics 107 (1):35–78.
- Li, Ivan. 2024. Name, Image, and Likeness in Marketing. Ph.D. thesis, Stanford University.
- Li, Ivan and Timothy Derdenger. 2024. "Do the Rich Get Richer? Competitiveness in the Student-Athlete Market under NIL." Mimeo.
- Liu, Haoming and Jingfeng Lu. 2006. "Measuring the Degree of Assortative Mating." Economics Letters 92:317–322.
- Mare, Robert D. 2001. "Observations on the Study of Social Mobility and Inequality." In Social Stratification: Class, Race, and Gender in Sociological Perspective, edited by David B. Grusky. Boulder, Colorado: Westview, 477–488.
- Mare, Robert D. and Christine R. Schwartz. 2005. "Trends in Educational Assortative Marriage from 1940 to 2003." *Demography* 42:621–646.
- Meyer, Margaret and Bruno Strulovici. 2012. "Increasing Interdependence of Multivariate Distributions." *Journal of Economic Theory* 147:1460–1489.

———. 2015. "Beyond Correlation: Measuring Independence through Complementarities." Mimeo.

- Robbins, Michael C. 2009. "A Measure of Social Homogamy." *American Anthropologist* 83:379–385.
- Romano, Joseph, Azeem Shaikh, and Michael Wolf. 2008. "Formalized Data Snooping Based on Generalized Error Rates." *Econometric Theory* 24:404–447.
- Romano, Joseph and Michael Wolf. 2005. "Stepwise Multiple Testing as Formalized Data Snooping." *Econometrica* 73:1237–1282.
- ———. 2016. "Efficient Computation of Adjusted P-Values for Resampling-Based Stepdown Multiple Testing." Statistics and Probability Letters 113:38–40.

Shen, Jenny. 2020. Essays in the Economics of Gender. Ph.D. thesis, Princeton University.

- Siow, Aloysius. 2015. "Testing Becker's Theory of Positive Assortative Matching." *Journal* of Labor Economics 33 (2).
- Wu, Jiabin and Hanzhe Zhang. 2021. "Preference Evolution in Different Matching Markets." European Economic Review 137:103804.
- Yule, G. Udny. 1900. "On the Association of Attributes in Statistics: With Illustrations from the Material of the Childhood Society." *Philosophical Transactions of the Royal Society* of London Series A 194:257–319.
- ———. 1912. "On the Methods of Measuring Association between Two Attributes." *Journal* of the Royal Statistical Society 75 (6):579–652.
- Zhang, Hanzhe. 2021. "An Investment-and-Marriage Model with Differential Fecundity: On the College Gender Gap." Journal of Political Economy 129 (5):1464–1486.

<sup>——. 2024. &</sup>quot;Axiomatic Measures of Assortative Matching." Mimeo.

## A Online Appendix

### A.1 Omitted proofs for characterization results

**Proof of Claim 2.** Suppose M = (a, b, c, d) and M' = (a', b', c', d') have the same marginal distributions and  $M \succ M'$ . By marginal independence,

$$\begin{aligned} (a,b,c,d) &\sim (a,b,c\frac{c'}{c},d\frac{c'}{c}) \sim (a,b\frac{c}{c'}\frac{d'}{d},c',d\frac{c'}{c}\frac{c}{c'}\frac{d'}{d}) \sim (a\frac{b'}{b}\frac{c'}{c}\frac{d}{d'},b\frac{b'}{b}\frac{c'}{c}\frac{d}{d'}\frac{c}{c'}\frac{d'}{d},c',d') \\ &\sim (a'\frac{a}{a'}\frac{b'}{b}\frac{c'}{c}\frac{d}{d'},b',c',d'). \end{aligned}$$

Let  $\delta \equiv \frac{a}{a'} \frac{b'}{b} \frac{c'}{c} \frac{d}{d'}$ . By marginal monotonicity, a > a', b < b', c > c', and d > d'. Hence,  $\delta > 1$ . The relations  $(a\delta, b', c', d') \sim (a', b'/\delta, c', d') \sim (a', b', c'/\delta, d') \sim (a', b', c', d'\delta) \succ (a', b', c', d')$  for all a', b', c', and d' imply diagonal monotonicity and off-diagonal monotonicity.

**Proof of Claim 3.** Suppose preorder  $\succeq$  satisfies marginal monotonicity. Then we have for any  $\delta$ ,

$$(a, b, c, d) \succ (a - \delta, b + \delta, c + \delta, d + \delta).$$

By (repeatedly applying) marginal independence,

$$\begin{aligned} (a - \delta, b + \delta, c + \delta, d + \delta) &\sim (a, b + \delta, (c + \delta) \frac{a}{a - \delta}, d - \delta) \\ &\sim (a, b, (c + \delta) \frac{a}{a - \delta}, (d - \delta) \frac{b}{b + \delta}) \\ &\sim (a, b, c, d \frac{a - \delta}{a} \frac{b}{b + \delta} \frac{c}{c + \delta} \frac{d - \delta}{d}) \equiv (a, b, c, d\lambda(\delta)), \end{aligned}$$

where  $\lambda(\delta)$  is strictly smaller than 1 for any  $\delta > 0$  and is continuously decreasing in  $\delta$ . With similar transformations, we have

$$(a - \delta, b + \delta, c + \delta, d + \delta)$$
  
~  $(a, b, c, d\lambda(\delta)) \sim (a\lambda(\delta), b, c, d) \sim (a, b/\lambda(\delta), c, d) \sim (a, b, c/\lambda(\delta), d)$ 

Hence, for any  $\delta > 0$ , diagonal monotonicity holds:

$$(a, b, c, d) \succ (a, b, c, d\lambda(\delta)) \sim (\lambda(\delta)a, b, c, d)$$

and off-diagonal monotonicity holds:

$$(a, b, c, d) \succ (a, b/\lambda(\delta), c, d) \sim (a, b, c/\lambda(\delta), d)$$

Reversely,

$$(a, b, c, d) \succ (a, b, c, d\lambda(\delta)) \sim (\lambda(\delta)a, b, c, d)$$

for any  $\delta$  implies

$$(a, b, c, d) \succ (a, b/\lambda(\delta), c, d) \sim (a, b, c/\lambda(\delta), d)$$

for any  $\delta$ , and

$$(a, b, c, d) \succ I(a - \delta, b + \delta, c + \delta, d - \delta)$$

for any  $\delta$ , where marginal independence is repeatedly used again.

**Proof of Theorem 1.** It is straightforward to check that the preorder induced by the odds ratio satisfies marginal monotonicity, maximum homogamy, and marginal independence.

It remains to show that there does not exist an index I (or equivalently in the current context, a preorder induced by index I) that satisfies these axioms but is not preorderequivalent to the odds ratio. Suppose by way of contradiction that such an index exists. Then, we must have for some M = (a, b, c, d) and M' = (a', b', c', d'). One of the following four cases occurs: (i) I(M) > I(M') and Q(M) < Q(M'), (ii) I(M) < I(M') and Q(M) > Q(M'), (iii) I(M) = I(M') and  $Q(M) \neq Q(M')$ , and (iv)  $I(M) \neq I(M')$  and Q(M) = Q(M').

First, suppose that case (i) I(M) > I(M') and Q(M) < Q(M') occurs. Because I(M) > I(M'), by the implication of DM,  $ad \neq 0$ , and by the implication of ODM,  $b'c' \neq 0$ , and because Q(M) < Q(M'), similarly,  $bc \neq 0$  and  $a'd' \neq 0$ . For each of the following steps, we

invoke a part of marginal independence:

$$\begin{split} I(a, b, c, d) &= I(\frac{b'}{b}a, \frac{b'}{b}b, c, d) \\ &= I(\frac{b'}{b}a\frac{a'}{a}\frac{b}{b'}, b', c\frac{a'}{a}\frac{b}{b'}, d) \\ &= I(a', b', c\frac{a'}{a}\frac{b}{b'} \cdot \frac{c'}{c}\frac{a}{a'}\frac{b'}{b}, d \cdot \frac{c'}{c}\frac{a}{a'}\frac{b'}{b}\frac{d'}{d'}) \\ &= I(a', b', c', d' \cdot \frac{ad}{bc} / \frac{a'd'}{b'c'}). \end{split}$$

By premise,

$$I(a',b',c',d'\cdot\frac{ad}{bc}/\frac{a'd'}{b'c'})>I(a',b',c',d').$$

By diagonal monotonicity, this implies

$$\frac{ad}{bc} > \frac{a'd'}{b'c'}$$

However, this implies Q(M) > Q(M'), which contradicts the premise that Q(M) < Q(M').

For each of the four possibilities, using the same logic as above, a contradiction can be derived for any positive matchings M and M'.

Suppose there is a cell that is zero. Suppose I(M) = I(M'). If bc = 0, then I(M) = I(a, 0, 0, d) = I(M'), then b'c' = 0. In this case, by DM, Q(M) = Q(M'). If ad = 0 instead, then I(M) = I(0, b, c, 0) = I(M') implies a'd' = 0. In this case, by ODM, Q(M) = Q(M'). Hence, whenever there is a cell with zero in one of the matrices, I(M) = I(M') and Q(M) = Q(M'), which prevents all four cases from happening.

**Proof of Claim 4.** Consider M = (a, b, c, d) > 0 and M' = (a', b, c, d) > 0, where a' > a. We want to show that I(M') > I(M). By type invariance,

$$I(a, b, c, d) = I(d, c, b, a).$$

By population decomposability,

$$\frac{1}{2}I(a,b,c,d) + \frac{1}{2}I(d,c,b,a) = I\left(\frac{1}{2}(a+d), \frac{1}{2}(b+c), \frac{1}{2}(b+c), \frac{1}{2}(a+d)\right).$$

By scale invariance,

$$I(a,b,c,d) = I\left(\frac{a+d}{a+b+c+d}, \frac{b+c}{a+b+c+d}, \frac{b+c}{a+b+c+d}, \frac{a+d}{a+b+c+d}\right).$$

By the same sequence of arguments by population decomposability, scale invariance, and type invariance,

$$I(a', b, c, d) = I\left(\frac{a'+d}{a'+b+c+d}, \frac{b+c}{a'+b+c+d}, \frac{b+c}{a'+b+c+d}, \frac{a'+d}{a'+b+c+d}\right).$$

Note that the two matrices on the right-hand side of the two equations above have the same marginals (each row or column sums to 1). By marginal monotonicity, a' > a implies

$$I(M) = I(a', b, c, d) > I(M') = I(a, b, c, d).$$

Hence, diagonal monotonicity is proved. Off-diagonal monotonicity can be shown analogously.  $\hfill \square$ 

**Proof of Theorem 2.** We first show that any index I that satisfies scale, side and type invariance, as well as diagonal and off-diagonal monotonicity, and population decomposability is order-equivalent to—i.e., a monotonic transformation of—normalized trace. Consider  $M = (a, b, c, d) > \mathbf{0}$  and  $M' = (a', b', c', d') > \mathbf{0}$ . Here, we prove a lemma.

**Lemma 1.** Any index that satisfies scale, side and type invariance, as well as diagonal and off-diagonal monotonicity, and population decomposability is an increasing function of a + d and a decreasing function of b + c.

**Proof of Lemma 1.** Consider M = (a, x, x, d), and consider  $M_1 = (a/2, x - \lambda/2, \lambda/2, d/2)$ 

and  $M_2 = (a/2, \lambda/2, x - \lambda/2, d/2)$  for some  $\lambda \in (0, 2x)$ . Note that  $M_1 + M_2 = M$  and  $M_1$ and  $M_2$  have the same total mass. Hence, by population decomposability,

$$I(M) = \frac{1}{2}I(M_1) + \frac{1}{2}(M_2).$$

By side invariance,  $I(M_1) = I(M_2)$ . Hence,  $I(M) = I(M_1) = I(M_2)$ . By scale invariance,

$$I(M_1) = I(2M_1) = I(a, 2x - \lambda, \lambda, d).$$

Hence, for all  $\lambda \in (0, 2x)$ ,

$$I(a, 2x - \lambda, \lambda, d) = I(a, x, x, d).$$

Hence, we have shown that I(a, b, c, d) = I(a, b', c', d) whenever b + c = b' + c'. By the same logic, and by side invariance and type invariance, I(a, b, c, d) = I(a', b, c, d') whenever a + d = a' + d'. I, by diagonal monotonicity, is strictly increasing in a + d whenever  $bc \neq 0$ , and, by off-diagonal monotonicity, is strictly decreasing in b + c whenever  $ad \neq 0$ .  $\Box$ 

If (i) a + d > a' + d' and  $b + c \le b' + c'$  or (ii)  $a + d \le a' + d'$  and b + c > b' + c', then, by Lemma 1, (i) I(M) > I(M') or (ii) I(M) < I(M'), respectively. Suppose a + d > a' + d'and b + c > b' + c'. Define

$$M'' = (a'', b'', c'', d'') = M' \cdot (b+c)/(b'+c').$$

By definition of M'', b'' + c'' = b + c, and  $a'' + d'' = (a' + d') \cdot (b + c)/(b' + c')$ . By scale invariance of I, I(M'') = I(M'). The comparison of a + d and a'' + d'' pins down the ordinal assortativeness relation between M and M'. That is,

$$\frac{a+d}{a''+d''} = \frac{a+d}{b+c} \bigg/ \frac{a'+d'}{b'+c'} > 1 \Leftrightarrow I(M) > I(M').$$

When a + d < a' + d' and b + c < b' + c', we can similarly pin down the ordinal assortativeness

relation between M and M'. Note that for any M = (a, b, c, d) > 0,

$$I_{tr}(M) = \frac{(a+d)}{(a+b) + (c+d)} = \frac{a+d}{b+c} \bigg/ \bigg( \frac{a+d}{b+c} + 1 \bigg).$$

Hence, I(M) > I(M') if and only if  $I_{tr}(M) > I_{tr}(M')$ .

Any nonlinear transformation of normalized trace would violate population decomposability, so any index that satisfies the stated axioms must be not only a monotonic transformation but also a positive affine transformation of normalized trace.  $\Box$ 

**Proof of Theorem 3.** It is straightforward to check that the aggregate likelihood ratio satisfies the axioms, so it remains to show the other direction. We first show that any index I that satisfies all three invariance axioms, as well as marginal monotonicity, and random decomposability is proportional to the aggregate likelihood ratio. Consider M = (a, b, c, d). By type invariance,

$$I(M) = I \begin{pmatrix} a & b \\ c & d \end{pmatrix} = I \begin{pmatrix} d & c \\ b & a \end{pmatrix}.$$
 (7)

Recall

$$r(M) \equiv \frac{a+b}{|M|} + \frac{a+c}{|M|} + \frac{d+b}{|M|}\frac{d+c}{|M|}|M| = \frac{(a+b)(a+c) + (d+b)(d+c)}{a+b+c+d}$$

By random decomposability,

$$I\begin{pmatrix}a+d & b+c\\b+c & a+d\end{pmatrix} \cdot r\begin{pmatrix}a+d & b+c\\b+c & a+d\end{pmatrix} = I\begin{pmatrix}a & b\\c & d\end{pmatrix} \cdot r\begin{pmatrix}a & b\\c & d\end{pmatrix} + I\begin{pmatrix}d & c\\b & a\end{pmatrix} \cdot r\begin{pmatrix}d & c\\b & a\end{pmatrix},$$

which, by (7), is simplified to

$$I\begin{pmatrix} a+d & b+c\\ b+c & a+d \end{pmatrix} \frac{|M|^2 + |M|^2}{2|M|} = 2I(M)r(M) \Rightarrow I(M) = \frac{1}{2}\frac{|M|}{r(M)}I\begin{pmatrix} a+d & b+c\\ b+c & a+d \end{pmatrix}.$$
 (8)

Because for any  $\epsilon < \min\{a+d, b+c\},\$ 

$$\begin{pmatrix} a+d & b+c \\ b+c & a+d \end{pmatrix} = \begin{pmatrix} a+d-\epsilon & \epsilon \\ \epsilon & a+d-\epsilon \end{pmatrix} + \begin{pmatrix} \epsilon & b+c-\epsilon \\ b+c-\epsilon & \epsilon \end{pmatrix},$$

by random decomposability, for any  $\epsilon < \min\{a + d, b + c\}$ ,

$$|M|I\begin{pmatrix}a+d&b+c\\b+c&a+d\end{pmatrix} = (a+d)I\begin{pmatrix}a+d-\epsilon&\epsilon\\\epsilon&a+d-\epsilon\end{pmatrix} + (b+c)I\begin{pmatrix}\epsilon&b+c-\epsilon\\b+c-\epsilon&\epsilon\end{pmatrix}$$

Plugging in the expression of I(M) in (8), we get, for any  $\epsilon < \min\{a + d, b + c\}$ ,

$$I(M) = \frac{1}{2r(M)} \left[ (a+d) \cdot I \begin{pmatrix} a+d-\epsilon & \epsilon \\ \epsilon & a+d-\epsilon \end{pmatrix} + (b+c) \cdot I \begin{pmatrix} \epsilon & b+c-\epsilon \\ b+c-\epsilon & \epsilon \end{pmatrix} \right].$$

Take  $\epsilon \to 0^+$  and by scale invariance and I(0, 1, 1, 0) = 0, we have

$$I(M) = \frac{1}{2} \frac{a+d}{r(M)} I \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}$$

Hence, any index that satisfies the above axioms is proportional to (a+d)/r(M), the aggregate likelihood ratio.

### A.2 Iso-assortative surfaces

Fix a reference matching, M = (0.3, 0.1, 0.1, 0.5) and consider markets with a unit mass of couples, |M'|=1. Figure 2a shows what we call *iso-assortative surfaces* in the (a, b, d) threedimensional space: The red iso-assortative surface collects the matching matrices (a, b, 1 - a - b - d, d) that are equally assortative to M according to the aggregate likelihood ratio, and the blue iso-assortative surface collects the matching matrices that are equally assortative to M according to the odds ratio. Because the iso-assortative surfaces are different, the



Figure 2: Iso-assortative surfaces and conflicting conclusions

matching matrices that are judged to be more assortative than M by the two measures will differ. The orange region in Figure 2b consists of the matching matrices that are more assortative than M under the aggregate likelihood ratio but are less assortative than M under the odds ratio. Figure 1 illustrates *iso-assortative curves* and conflicting regions between the aggregate likelihood ratio and the odds ratio in a plane, by restricting the attention to matching matrices in which b = c. Mathematically speaking, Figure 1 illustrates the sliced plane of b = (1 - a - d)/2 in the (a, b, d) space of Figures 2a and 2b.

### A.3 Supplementary empirical results

	Submarkets			Global indices		
	C&PG vs SC	C&PG vs SC PG vs C C vs SC C&PG v		C&PG vs others	5  educ levels	
	(1)	(2)	(3)	(4)	(5)	
	Panel A: Cohorts 1970s vs 1930s					
Odds ratio	2.681	-1.965	1.912	-6.252		
	[00.] (00.)	[00.]	[00.]	(.00) $[.00]$		
L1 ratio (high ed)	0.071	-0.290	0.163	-1.889		
	(.00) $[.00]$	[00.] (00.)	[00.]	(.00) $[.00]$		
L2 ratio (low ed)	0.299	0.063	0.092	0.268		
	(.00) $[.00]$	[00.]	(.00) $[.00]$	(.00) $[.00]$		
Weighted L ratio	0.116	0.043	0.133	0.327	0.383	
	(.00) $[.00]$	(.00) $[.00]$	(.00) $[.00]$	(.00) $[.00]$	[00.] (00.)	
Normalized trace	0.102	0.011	0.084	-0.083	-0.028	
0	(.00) $[.00]$	(.23) [.14]	(.00) $[.00]$	(.00) $[.00]$	(.00) $[.00]$	
$\chi^2$	0.097	-0.010	0.080	0.060		
	(.00) [.00]	(.13) [.13]	(.00) [.00]	(.00) [.00]		
Minimum distance	0.022	-0.300	0.044	-0.080		
	(.20) [.10]	(.00) [.00]	(.01) $[.00]$	(.00) [.00]		
	1.070	Panel I	B: Cohorts I	.970s vs 1940s		
Odds ratio	1.972	-0.852	1.597	-2.384		
	(.00) [.00]	(.00) [.00]	(.00) [.00]	(.00) [.00]		
L1 ratio (high ed)	0.048	-0.075	0.109	-0.765		
	(.00) [.00]	(.00) [.00]	(.00) [.00]	(.00) [.00]		
L2 ratio (low ed)	0.198		0.091	(00) [00]		
Weighted I metic	(.00) [.00]	(.98) [.98]	(.00) [.00]	(.00) $[.00]$	0.956	
weighted L ratio	870.0 [00.] (00.)	-0.014	001.0	(0.195)	062.0	
Normalized trace	(.00) $[.00]$	(.32) $[.20]$	(.00) $[.00]$	(.00) $[.00]$	(.00) $[.00]$	
Normalized trace	(00) [00]	(0.010)	(00) [00]	100.0-	(24) [08]	
$\lambda^2$	(.00) [.00]	(.01) $[.00]$	0.064	(.00) $[.00]$	(.24) [.06]	
X	(00)	(06) [03]	(00) [00]	(0.028)		
Minimum distance	0.003	0 101	(.00) $[.00]$	(.00) [.00]		
Willing distance	(03)[77]	(00)[00]	(0.055)	(00) [00]		
	(.55) [.11]	(.00) [.00] Panel (	C Cohorts 1	970s vs 1960s		
Odds ratio	0.277	-0.170	0.236	0.279		
0 4 40 1 4 10	(.48) [.16]	(.48) [.18]	(.53) [.20]	(.69) [.27]		
L1 ratio (high ed)	-0.054	-0.078	-0.042	-0.270		
	[00.] (00.)	[00.] (00.)	[00.] (00.)	[00.] (00.)		
L2 ratio (low ed)	0.152	0.026	0.058	0.105		
()	[00.] (00.)	[00.] (00.)	[00.] (00.)	[00.] (00.)		
Weighted L ratio	-0.039	0.013	0.014	0.087	0.130	
0	[00.] (00.)	(.49) $[.17]$	(.32) $[.10]$	[00.]	[00.] (00.)	
Normalized trace	0.015	-0.012	0.003	-0.015	-0.002	
	[00.] (00.)	(.05) $[.01]$	(.79) [.39]	(.00) [.00]	(.87) [.52]	
$\chi^2$	-0.001	-0.002	0.008	0.016		
	(.88) [.88]	(.87) [.65]	(.52) $[.20]$	(.01) [.00]		
Minimum distance	0.012	-0.042	0.026	0.068		
	(.49) [.20]	(.00) $[.00]$	(.05) $[.02]$	(.00) [.00]		

Table 5: Marital assortativeness: comparing cohort born 1970-75 with earlier cohorts

Notes: Education in 5 groups: post-graduate degrees (PG), 4-years college degrees (C), some college (SC), high school qualifications (HS), no qualifications (HSD). Column 1 refers to matching matrices between PG and C together and SC; column 2: PG and C; column 3: C and SC; column 4: PG and C together versus others; column 5: all 5 groups separate. For each index, top row shows estimates of the difference between the latest and earliest cohorts; the number in round brackets shows *p*-values for 2-sided significance testing adjusted for multiple hypothesis using the stepdown method jointly for the 30 measures on the panel (Romano and Wolf, 2005; Romano, Shaikh, and Wolf, 2008; Romano and Wolf, 2016); the number in square brackets shows stepdown *p*-values but for each market (column) separately. *Data source*: March extract of the US CPS, subsample of legally married and cohabiting individuals observed aged 35-44 in birth cohorts 1930-39, 1940-49, 1950-59, 1960-69 and 1970-75.